# UC San Diego UC San Diego Previously Published Works

# Title

Adipose tissue biomarkers and type 2 diabetes incidence in normoglycemic participants in the MESArthritis Ancillary Study: A cohort study

**Permalink** https://escholarship.org/uc/item/7083c5pc

**Journal** PLOS Medicine, 18(7)

**ISSN** 1549-1277

## **Authors**

Pishgar, Farhad Shabani, Mahsima Silva, Thiago Quinaglia AC <u>et al.</u>

**Publication Date** 

2021

# DOI

10.1371/journal.pmed.1003700

Peer reviewed



# 

**Citation:** Pishgar F, Shabani M, Quinaglia A. C. Silva T, Bluemke DA, Budoff M, Barr RG, et al. (2021) Adipose tissue biomarkers and type 2 diabetes incidence in normoglycemic participants in the MESArthritis Ancillary Study: A cohort study. PLoS Med 18(7): e1003700. https://doi.org/ 10.1371/journal.pmed.1003700

Academic Editor: Weiping Jia, Shanghai Jiao Tong University Affiliated Sixth People's Hospital, CHINA

Received: January 22, 2021

Accepted: June 16, 2021

Published: July 9, 2021

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1003700

**Copyright:** © 2021 Pishgar et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** Data are available at MESA-NHLBI.org, upon receiving approval from the Multi-Ethnic Study of Atherosclerosis (MESA)

**RESEARCH ARTICLE** 

Adipose tissue biomarkers and type 2 diabetes incidence in normoglycemic participants in the MESArthritis Ancillary Study: A cohort study

Farhad Pishgar<sup>1</sup>, Mahsima Shabani<sup>2</sup>, Thiago Quinaglia A. C. Silva<sup>2</sup>, David A. Bluemke<sup>3</sup>, Matthew Budoff<sup>4</sup>, R Graham Barr<sup>5,6</sup>, Matthew A. Allison<sup>7</sup>, Alain G. Bertoni<sup>8</sup>, Wendy S. Post<sup>2</sup>, João A. C. Lima<sup>2</sup>, Shadpour Demehri<sup>1\*</sup>

1 Russell H. Morgan Department of Radiology and Radiological Science, Johns Hopkins University School of Medicine, Baltimore, Maryland, United States of America, 2 Division of Cardiology, Department of Medicine, Johns Hopkins University School of Medicine, Baltimore, Maryland, United States of America, 3 Department of Radiology, University of Wisconsin School of Medicine and Public Health, Madison, Wisconsin, United States of America, 4 Lundquist Institute at Harbor-University of California Los Angeles School of Medicine, Torrance, California, United States of America, 5 Department of Medicine, Columbia University Medical Center, New York, New York, United States of America, 6 Department of Epidemiology, Columbia University Medical Center, New York, New York, United States of America, 7 Division of Preventive Medicine, Department of Family Medicine and Public Health, University of California San Diego, La Jolla, California, United States of America, 8 Division of Public Health Sciences, Wake Forest School of Medicine, Winston-Salem, North Carolina, United States of America

\* Demehri2001@Yahoo.com, SDemehr1@JHMI.edu

# Abstract

## Background

Given the central role of skeletal muscles in glucose homeostasis, deposition of adipose depots beneath the fascia of muscles (versus subcutaneous adipose tissue [SAT]) may precede insulin resistance and type 2 diabetes (T2D) incidence. This study was aimed to investigate the associations between computed tomography (CT)–derived biomarkers for adipose tissue and T2D incidence in normoglycemic adults.

## Methods and findings

This study was a population-based multiethnic retrospective cohort of 1,744 participants in the Multi-Ethnic Study of Atherosclerosis (MESA) with normoglycemia (baseline fasting plasma glucose [FPG] less than 100 mg/dL) from 6 United States of America communities. Participants were followed from April 2010 and January 2012 to December 2017, for a median of 7 years. The intermuscular adipose tissue (IMAT) and SAT areas were measured in baseline chest CT exams and were corrected by height squared (SAT and IMAT indices) using a predefined measurement protocol. T2D incidence, as the main outcome, was based on follow-up FPG, review of hospital records, or self-reported physician diagnoses.

Participants' mean age was  $69 \pm 9$  years at baseline, and 977 (56.0%) were women. Over a median of 7 years, 103 (5.9%) participants were diagnosed with T2D, and 147 (8.4%) participants died. The IMAT index (hazard ratio [HR]: 1.27 [95% confidence interval steering committee and local institutional ethics board(s). Please see MESA-NHLBI.org/ Publications.aspx or email MESA Coordinating Center at chsccweb@u.washington.edu for more details.

Funding: The MESA study was supported by contracts 75N92020D00001, HHSN268201500003I, N01-HC-95159, 75N92020D00005, N01-HC-95160, 75N92020D00002. N01-HC-95161. 75N92020D00003, N01-HC-95162, 75N92020D00006, N01-HC-95163, 75N92020D00004, N01-HC-95164, 75N92020D00007, N01-HC-95165, N01-HC-95166, N01-HC-95167, N01-HC-95168, N01-HC-95169. R01-HL-077612. and R01-HL-093081 from the National Heart, Lung, and Blood Institute (NHLBI), and by grants UL1-TR-000040, UL1-TR-001079, and UL1-TR-001420 from the National Center for Advancing Translational Sciences (NCATS). This publication was developed under a STAR research assistance agreement. No. RD831697 (MESA Air) and RD-83830001 (MESA Air Next Stage), awarded by the U.S Environmental Protection Agency (EPA). It has not been formally reviewed by the EPA. The views expressed in this document are solely those of the authors and the EPA does not endorse any products or commercial services mentioned in this publication. The aforementioned funding agencies provided financial support of the study for participant recruitment and data collection (paid to the MESA field centers). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. The specific roles of the authors affiliated with the MESA field centers are articulated in the 'author contributions' section.

**Competing interests:** The authors have read the journal's policy and have the following competing interest(s): Relevant to this study, RGB received grant funding from the National Institutes of Health (NIH) and the Chronic Obstructive Pulmonary Disease (COPD) Foundation. Other authors have declared that no competing interests exist.

Abbreviations: ASCVD, atherosclerotic cardiovascular disease; BMI, body mass index; CI, confidence interval; CT, computed tomography; FPG, fasting plasma glucose; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; HU, Hounsfield unit; IMAT, intermuscular adipose tissue; MESA, Multi-Ethnic Study of Atherosclerosis; MET, metabolic equivalent; PM, pectoralis muscle; SAT, subcutaneous adipose tissue; SD, standard deviation; STROBE, [CI]: 1.15–1.41] per 1-standard deviation [SD] increment) and the SAT index (HR: 1.43 [95% CI: 1.16–1.77] per 1-SD increment) at baseline were associated with T2D incidence over the follow-up. The associations of the IMAT and SAT indices with T2D incidence were attenuated after adjustment for body mass index (BMI) and waist circumference, with HRs of 1.23 (95% CI: 1.09–1.38) and 1.29 (95% CI: 0.96–1.74) per 1-SD increment, respectively. The limitations of this study include unmeasured residual confounders and one-time measurement of adipose tissue biomarkers.

## Conclusions

In this study, we observed an association between IMAT at baseline and T2D incidence over the follow-up. This study suggests the potential role of intermuscular adipose depots in the pathophysiology of T2D.

## **Trial registration**

ClinicalTrials.gov NCT00005487

## Author summary

## Why was this study done?

• This study was designed to investigate the associations between computed tomography (CT)-derived biomarkers for adipose tissue at baseline and type 2 diabetes (T2D) incidence in normoglycemic adults.

## What did the researchers do and find?

- We assessed 1,744 normoglycemic participants with chest CT exams between 2010 and 2012. The intermuscular adipose tissue (IMAT) and subcutaneous adipose tissue (SAT) areas were measured in these chest CT exams using a predefined measurement protocol.
- Participants were followed from April 2010 and January 2012 to December 2017, for a median of 7 years, for T2D incidence. T2D incidence was based on follow-up fasting plasma glucose (FPG), review of hospital records, or self-reported physician diagnoses.
- This study found that a higher CT-derived IMAT at baseline was associated with T2D incidence over the follow-up.

## What do these findings mean?

• In normoglycemic participants, the IMAT deposition was associated with T2D incidence. This study suggests the potential role of intermuscular adipose depots in the pathophysiology of T2D.

Strengthening the Reporting of Observational Studies in Epidemiology; T2D, type 2 diabetes.

• The CT-derived adipose tissue biomarkers are obtainable from CT exams performed for other initial indications and can extend the value of the routinely performed chest CT exams. Such biomarkers may be associated with T2D incidence.

#### Introduction

Type 2 diabetes (T2D) affects more than 20 million new cases every year, and, in addition to the attributable high morbidity and mortality, imposes an increased financial burden on healthcare systems worldwide [1]. However, most of this burden can be avoided by early identification of at-risk individuals and the rapid implementation of primary and secondary preventive measures [2]. Therefore, identifying the biomarkers associated with T2D incidence is crucial to minimize the morbidity, mortality, and healthcare financial burden attributable to this prevalent and chronic disease.

There are known associations between obesity or excessive overall adipose depots and the T2D incidence [3–5]. However, recent data from genome-wide association studies [6,7] and imaging assessments [7] suggest a stronger association between ectopic adipose depots (in liver, visceral organs, and muscles) and T2D incidence (compared to excessive overall adipose depots) [8–10]. Specifically, adipose depots deposition beneath the fascia of skeletal muscles (extramyocellular and intramyocellular lipid content) may contribute to T2D incidence, owing to the skeletal muscles cardinal role in glucose homeostasis [11]. The body mass index (BMI) and other clinical anthropometric indices, despite quantifying the excessive overall adipose depots, do not provide data on ectopic adipose depots distribution.

Imaging-derived adipose tissue biomarkers have the potential to provide a better characterization of ectopic adipose depots distribution [12]. In addition to dedicated imaging techniques for evaluating body composition and adipose depots distribution (e.g., dual-energy Xray absorptiometry or bioelectrical impedance), chest computed tomography (CT) exams that are commonly performed in the routine clinical practice for cardiopulmonary indications (e.g., coronary calcium scoring [13] or lung cancer screening [14]) retain data on adipose depots distribution, and there is an opportunity to extract biomarkers on the distribution of ectopic adipose depots from these CT exams, at zero additional cost or radiation exposure [15–17].

To our knowledge, no prior studies assessed the associations between CT-derived adipose depots biomarkers and T2D incidence using longitudinal analysis [3,8–10] after adjusting for the effects of traditional risk factors of the disease and clinical anthropometric indices (e.g., BMI and waist circumference) [8,9] or stratified for baseline glycemic status (normoglycemia versus prediabetes) [9].

In the present study, we used the population-based multiethnic cohort of participants in the Multi-Ethnic Study of Atherosclerosis (MESA) to assess the associations between CT-derived adipose depots biomarkers and T2D incidence. We characterized intermuscular adipose tissue (IMAT), subcutaneous adipose tissue (SAT), and intramyocellular lipid contents in the base-line chest CT exams and studied the potential associations between these biomarkers and T2D incidence.

#### Methods

The MESA is a population-based multiethnic cohort of 6,814 participants from 6 communities across the USA to investigate the features of subclinical and clinical cardiovascular diseases

and to determine their relevant risk factors (see www.MESA-NHLBI.org) [18]. The MESA was approved by the institutional review boards of the 6 participating field centers (Columbia University, Johns Hopkins University, Northwestern University, University of California, University of Minnesota, and Wake Forest University) and the coordinating center (University of Washington). All participants in the MESA provided written informed consent (registered at ClinicalTrials.gov as NCT00005487).

#### **MESArthritis Ancillary Study**

Between April 2010 and January 2012 (fifth MESA exam), 3,137 participants in the MESA consented to participate in the MESA Lung Ancillary Study and underwent chest CT exams [19]. The MESArthritis Ancillary Study is an analysis of the available CT exams of 3,083 of these participants [20]. It is aimed to investigate the roles of CT-derived soft tissue and bone biomarkers associated with incidence and clinical outcomes of several cardiometabolic and cardiopulmonary diseases. Before this study, 86 participants with low-quality chest CT exams (n = 52), unknown baseline glycemic status (n = 21), or missing follow-up information (n = 13) were excluded (S1 Fig).

In the baseline exam, information on various demographic and clinical characteristics was collected. Participants were visited to obtain plasma samples after 12 hours of fasting to measure plasma glucose, insulin, HbA1c, triglyceride, and high-density lipoprotein (HDL) cholesterol [21,22]. In this study, normoglycemic participants (baseline fasting plasma glucose [FPG] less than 100 mg/dL) were included (S1 Fig; participants with prediabetes, FPG of 100 to 125 mg/dL, were included in a supplementary analysis).

### **T2D diagnosis**

Between baseline (April 2010 to January 2012) to December 2017, participants were contacted by interviewers at intervals of 9 to 12 months to inquire about new disease diagnoses and interim hospital admissions. Moreover, between September 2016 and June 2018, participants completed a follow-up exam to recollect information on clinical characteristics and obtain fast-ing plasma samples (sixth MESA exam). This information was supplemented by the data collected through reviews of the hospital records (S2 Fig).

T2D diagnosis over the follow-up was based on at least 1 of the following criteria: (1) physician-diagnosed T2D based on the review of the hospital records (based on the relevant codes of the ninth and 10th editions of the International Classification of Diseases, S1 Table, time of incidence was the time of hospital admission); (2) self-reported physician-diagnosed T2D (time of incidence was the midpoint between the last interview without and the interview with the self-reported physician-diagnosed disease); or (3) use of insulin or oral hypoglycemic agents or FPG  $\geq$ 126 mg/dL in the follow-up exam (time of incidence was the time of the follow-up exam).

In a sensitivity analysis, self-reported physician-diagnosed T2D (the second criterion) was confirmed with the use of insulin or oral hypoglycemic agents or FPG  $\geq$ 126 mg/dL in the follow-up exam, and participants with self-reported T2D but missing information on the use of insulin or oral hypoglycemic agents or FPG in the follow-up exam (n = 17) were excluded.

#### Adipose tissue biomarkers

The non-contrast–enhanced chest CT exams (acquired at suspended full inspiration using 64-slice multidetector row CT scanners, Siemens Medical Solutions, Erlangen, Germany or GE Healthcare, Waukesha, Wisconsin, USA) were used to measure the IMAT and SAT areas as well as the pectoralis muscles (PMs) density [19,23]. The cross-sectional SAT area was

measured as the area between the PM and skin surface in the slice just above the superior margin of the aortic arch [20,24]. Density of the SAT area was analyzed to estimate an individualized attenuation threshold for the IMAT (i.e., the extramyocellular lipid content) [20,25]. The cross-sectional areas within the PM with attenuation below the estimated threshold were measured as the IMAT (S3 Fig). The IMAT and SAT areas were corrected as IMAT and SAT indices (area by height squared, cm<sup>2</sup>/m<sup>2</sup>) to account for the anthropometric variations. Moreover, the mean density of the PM area (after excluding the IMAT area) was also measured as a surrogate measure of the intramyocellular lipid content, Hounsfield unit by area, HU/cm<sup>2</sup>, S3 Fig).

#### Statistical analysis

This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guidelines (S1 Checklist). The statistical analyses of this study were planned a priori at the time of preparing the research proposal and was approved by the MESA publication and steering committees. Few additions (e.g., assessing the linearity of the associations between adipose tissue biomarkers and T2D incidence and multiply imputing the missing data points) or changes (e.g., reporting the associations per 1-standard deviation [SD] increment in the biomarkers) were made to the planned statistical analyses at the time of data analysis or per peer reviewers comments (S2 Table).

Descriptive statistics were used to compare baseline demographic and clinical characteristics between normoglycemic participants with and without T2D incidence over the follow-up.

The correlations between adipose tissue biomarkers with baseline BMI, waist circumference, triglyceride, HDL cholesterol, FPG, insulin, HbA1c, and homeostatic model assessmentinsulin resistance (HOMA-IR, FPG × insulin/405) were analyzed using the Pearson correlation analysis and illustrated using a symmetric correlation matrix.

The generalized additive Cox proportional hazard models with integrated smoothness estimation were used to assess and illustrate the linearity of the associations between adipose tissue biomarkers and T2D incidence. Moreover, several adjusted Cox proportional hazard models were used to estimate the hazard ratios (HRs) and 95% confidence interval (CI) for the T2D incidence according to adipose tissue biomarkers (per 1-SD increment). Models were tested for the proportional hazard assumption by regressing the Schoenfeld residuals over time. Models were adjusted for the traditional risk factors of T2D, HOMA-IR, and clinical anthropometric indices (i.e., BMI and waist circumference) at baseline. The traditional risk factors of T2D included age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity (vigorous and moderate metabolic equivalents [METs]), HDL cholesterol, triglyceride, and hypertension [3]. Decedents without T2D incidence were right-censored at the time of death.

Stratified analyses for the traditional risk factors of T2D, HOMA-IR, BMI, and waist circumference were conducted. In the HOMA-IR stratified analyses, the mean HOMA-IR in normoglycemic participants without T2D incidence over the follow-up was used as the cut point. High HDL cholesterol, high triglyceride, hypertension, and central obesity were defined according to the updated metabolic syndrome guideline of the National Cholesterol Education Program Adult Treatment Panel III [26]. Heterogeneity of the association between adipose tissue biomarkers and T2D incidence in the levels of the stratification variable was tested using the significance of a multiplicative interaction term between adipose tissue biomarkers and the stratification variable.

In a sensitivity analysis, self-reported physician-diagnosed T2D (the second criterion) was confirmed with the use of insulin or oral hypoglycemic agents or FPG  $\geq$ 126 mg/dL in the follow-up exam, and similar Cox proportional hazard models were used to study the associations between adipose tissue biomarkers and the T2D incidence.

In a supplementary analysis, similar Cox proportional hazard models were used to study the associations between adipose tissue biomarkers and the T2D incidence in participants with prediabetes at baseline.

The missing data points were multiply-imputed with chained equations and predictive mean matching method before Cox proportional hazard modeling and the stratified analyses to produce 5 datasets [27] Each dataset was analyzed separately, and the results were pooled across the datasets using Rubin's rule (missing values were infrequent among the traditional risk factors of T2D, S4 Fig).

We applied the Benjamini–Hochberg procedure to correct the *p*-values for multiple comparisons. The *p*-values from main, sensitivity, and supplementary analyses were batched and corrected separately using this procedure [28]. The associations with *p*-values <0.05 were considered statistically significant. All analyses were performed in the R platform, version 3.6.1 (R Foundation for Statistical Computing, Vienna, Austria).

#### Results

#### **Baseline characteristics**

Out of the 3,083 participants in the MESArthritis Ancillary Study, 1,744 normoglycemic participants at baseline were included in this study (S1 Fig). Participants' mean age was  $68.7 \pm 9.3$ years at baseline, and 977 (56.0%) participants were female (Table 1). Participants were followed for a median of 6.8 [6.2 to 7.2] years, and during this period, 103 (5.9%) participants were diagnosed with T2D, and 147 (8.4%) participants died (137 [7.9%] without T2D, Fig 1).

#### Intermuscular adiposity and T2D incidence

IMAT index correlated with BMI (r: 0.41, *p*-value: <0.001), waist circumference (r: 0.35, *p*-value: <0.001), HOMA-IR (r: 0.21, *p*-value: <0.001), and the HDL cholesterol (r: -0.15, *p*-value: <0.001) at baseline (S5 Fig).

The generalized additive Cox proportional hazard models showed little evidence for a nonlinear association between the IMAT index and T2D incidence (S6 Fig). In the models adjusted for traditional risk factors of T2D, higher IMAT index quartiles were associated with T2D incidence, and 1-SD increment in the IMAT index was associated with the disease incidence (HR: 1.27 [95% CI: 1.15 to 1.41], *p*-value <0.001, per 1-SD increment). This association was attenuated but remained statistically significant after adjusting for the effects of the HOMA-IR (HR: 1.26 [95% CI: 1.13 to 1.40], *p*-value: <0.001, per 1-SD increment) or BMI and waist circumference (HR: 1.23 [95% CI: 1.09 to 1.38], *p*-value: 0.010, per 1-SD increment) (Table 2).

In the stratified analyses, our models showed similar associations between IMAT index and T2D incidence in the strata of the traditional risk factors of T2D, HOMA-IR, BMI, and waist circumference (Fig 2, S7 Fig).

#### Subcutaneous adiposity and T2D incidence

The SAT index correlated with BMI (r: 0.65, *p*-value: <0.001), waist circumference (r: 0.46, *p*-value: <0.001), and HOMA-IR (r: 0.34, *p*-value: <0.001) at baseline (S5 Fig).

The generalized additive Cox proportional hazard models showed little evidence for a nonlinear association between the SAT index and T2D incidence (**S6 Fig**). Higher SAT index quartiles were associated with T2D incidence, and 1-SD increment in the SAT index was associated with T2D incidence (HR: 1.43 [95% CI: 1.16 to 1.77], *p*-value: 0.010, per 1-SD increment) after adjusting for traditional risk factors of T2D. After including the HOMA-IR (HR:

Characteristics	Without T2D incidence over the follow- up $(n = 1,641)$	With T2D incidence over the follow- up $(n = 103)$		
Traditional risk factors of T2D	-	· · · · ·		
Age (years)	68.7 ± 9.3	68.7 ± 9.5		
Sex (female)	924 (56.3%)	53 (51.5%)		
Race/ethnicity				
White	750 (45.7%)	35 (34.0%)		
Black	418 (25.5%)	30 (29.1%)		
Hispanic	279 (17.0%)	28 (27.2%)		
Chinese-American	194 (11.8%)	10 (9.7%)		
Physical activity (METs hours/week)	88.6 ± 96.5	85.1 ± 92.1		
Alcohol drinking status (current)	768 (46.8%)	32 (31.1%)		
Smoking status				
Never	757 (46.1%)	42 (40.8%)		
Former	747 (45.5%)	51 (49.5%)		
Current	127 (7.7%)	10 (9.7%)		
Systolic blood pressure (mm Hg)	122.2 ± 19.8	125.6 ± 21.3		
Diastolic blood pressure (mm Hg)	68.4 ± 9.9	70.0 ± 9.2		
TG (mg/dL)	$100.8 \pm 52.5$	$118.4 \pm 64.7$		
HDL cholesterol (mg/dL)	58.5 ± 17.2	52.4 ± 15.2		
HOMA-IR				
log <sub>2</sub> (HOMA-IR)	3.2 ± 0.9	3.6 ± 1.0		
Clinical anthropometric indices				
BMI (kg/m <sup>2</sup> )	27.4 ± 5.1	29.7 ± 5.6		
Waist C. (cm)	96.1 ± 13.4	$100.7 \pm 13.0$		
Adipose tissue biomarkers				
IMAT index (cm <sup>2</sup> /m <sup>2</sup> )	0.3 ± 0.3	$0.5 \pm 0.7$		
SAT index (cm <sup>2</sup> /m <sup>2</sup> )	18.8 ± 11.2	22.5 ± 14.6		
PM density (HU/cm <sup>2</sup> )	24.0 ± 10.3	23.4 ± 11.1		

Table 1. Baseline demographic and clinical characteristics of normoglycemic participants.

Quantitative variables are shown in mean ± SD, and qualitative variables are shown in number (%). BMI, body mass index; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment-insulin resistance; HU, Hounsfield unit; IMAT, intermuscular adipose tissue; IQR, interquartile range; MET, metabolic equivalent; PM, pectoralis muscle; SAT, subcutaneous adipose tissue; SD, standard deviation; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference.

https://doi.org/10.1371/journal.pmed.1003700.t001

1.34 [95% CI: 1.07 to 1.68], *p*-value: 0.078, per 1-SD increment) or BMI and waist circumference (HR: 1.29 [95% CI: 0.96 to 1.74], *p*-value: 0.325, per 1-SD increment), the associations between the SAT index and T2D incidence were attenuated toward null hypothesis and no longer were statistically significant (Table 2).

In the stratified analyses, our models showed similar associations between SAT index and T2D incidence in the strata of the traditional risk factors of T2D, HOMA-IR, BMI, and waist circumference (Fig 2, S8 Fig).



Fig 1. Flow diagram of participants and the timing of T2D diagnoses. \*Self-reported physician-diagnosed T2D. <sup>†</sup>Decedents without T2D at the time of death. FPG, fasting plasma glucose; T2D, type 2 diabetes.

https://doi.org/10.1371/journal.pmed.1003700.g001

#### Intramuscular adiposity and T2D incidence

The PM density correlated with BMI (r: -0.33, *p*-value: <0.001), waist circumference (r: -0.28, *p*-value: <0.001), and HOMA-IR (r: -0.16, *p*-value: <0.001) at baseline (S5 Fig).

The generalized additive Cox proportional hazard models showed little evidence for a nonlinear association between PM density and T2D incidence (S6 Fig). The models adjusted for

#### Table 2. Associations of adipose tissue biomarkers and T2D incidence.

	Index				<i>p</i> -value for trend	HR (95% CI), p-value per 1-SD increment		
	Quartile 1	Quartile 2	Quartile 3	Quartile 4				
IMAT index								
Mean (cm <sup>2</sup> /m <sup>2</sup> )	0.1	0.2	0.3	0.7	-	-		
Incident cases	16	19	26	42	-	-		
Incidence rate (per 1,000 PYs)	5.9	6.9	9.6	15.6	-	-		
HR (95% CI)								
Model 0	1 (reference)	1.18 (0.60-2.31)	1.62 (0.86-3.04)	2.66 (1.49-4.77)	< 0.001	1.27 (1.17–1.38), <0.001		
Model 1	1 (reference)	1.11 (0.55-2.25)	1.37 (0.70-2.69)	1.93 (1.03-3.64)	0.100	1.27 (1.15–1.41), <0.001		
Model 1 + HOMA-IR	1 (reference)	1.08 (0.54-2.18)	1.24 (0.63-2.45)	1.72 (0.91-3.26)	0.216	1.26 (1.13–1.40), <0.001		
Model 1 + BMI and Waist	1 (reference)	0.98 (0.48-1.99)	1.13 (0.56-2.27)	1.42 (0.71-2.86)	0.485	1.23 (1.09–1.38), 0.010		
<u>C.</u>								
SAT index								
Mean (cm <sup>2</sup> /m <sup>2</sup> )	7.6	13.1	20.4	35.1	-	-		
Incident cases	19	22	26	36	-	-		
Incidence rate (per 1,000 PYs)	7.3	8.0	9.3	13.2	-	-		
HR (95% CI)								
Model 0	1 (reference)	1.09 (0.58-2.02)	1.25 (0.69-2.28)	1.76 (1.00-3.09)	0.150	1.27 (1.08–1.50), 0.043		
Model 1	1 (reference)	1.09 (0.57-2.07)	1.53 (0.78-3.00)	2.65 (1.22-5.77)	0.078	1.43 (1.16–1.77), 0.010		
Model 1 + HOMA-IR	1 (reference)	1.00 (0.53-1.91)	1.26 (0.62-2.52)	2.02 (0.90-4.54)	0.300	1.34 (1.07–1.68), 0.078		
Model 1 + BMI and Waist C.	1 (reference)	0.98 (0.50-1.90)	1.26 (0.59–2.69)	1.79 (0.68–4.71)	0.495	1.29 (0.96–1.74), 0.325		

Model 0: unadjusted.

Model 1: adjusted for categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension. Reported *p*-values were corrected for multiple comparisons.

BMI, body mass index; CI, confidence interval; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; HU, Hounsfield unit; IMAT, intermuscular adipose tissue; PY, person-year; SAT, subcutaneous adipose tissue; SD, standard deviation; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference.

https://doi.org/10.1371/journal.pmed.1003700.t002

			IMAT Index		SAT Index		
Characteristics	Strata	Subjects, n	HR (95% CI)	p-value*	HR (95% CI)	p-value*	
Age	<60 years	339	1.00 (0.63 - 1.60)	0.585	1.42 (0.84 - 2.40)	0.924	اا ⊭⊕
	≥60 years	1405	1.31 (1.18 - 1.45)		1.39 (1.07 - 1.79)		⊨
Sex	Female	977	1.28 (1.11 - 1.49)	0.910	1.43 (1.11 - 1.83)	0.916	⊨- <b>□</b>   ⊨ <b>0</b>
	Male	767	1.26 (1.07 - 1.47)		1.45 (0.76 - 2.77)		►
Race/Ethnicity	White	785	1.26 (0.77 - 2.07)	0.910	1.27 (0.74 - 2.17)	0.565	►
	Other	959	1.26 (1.14 - 1.39)		1.46 (1.17 - 1.81)		⊨ <del>_ </del> + + ⊨ <b>-</b> 1
HOMA-IR	Low	1080	1.31 (1.13 - 1.52)	0.910	1.15 (0.72 - 1.83)	0.378	►
	High	664	1.20 (1.03 - 1.41)		1.43 (1.09 - 1.87)		⊨- <b>-</b>
BMI	≤25 kg/m2	608	0.97 (0.33 - 2.81)	0.909	1.45 (0.45 - 4.69)	0.909	¢
	>25 kg/m2	1136	1.26 (1.13 - 1.40)		1.35 (1.06 - 1.72)		►- <b>-</b>
Waist C.	Normal	858	1.74 (1.34 - 2.24)	0.060	0.83 (0.32 - 2.19)	0.341	€ <b>-</b> 1
	Central Obese	886	1.17 (1.03 - 1.33)		1.31 (1.01 - 1.69)		►
							(
							0.50 0.71 1.0 1.41 4.0 HR (95% CI)

📕 IMAT Index 🛑 SAT Index

**Fig 2.** Forest plot of the associations of adipose tissue biomarkers and T2D incidence by stratification variables. \**p*-value for interaction. Models were adjusted for covariates in Model 1 (categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension), except for the stratification variable. Reported *p*-values were corrected for multiple comparisons. BMI, body mass index; CI, confidence interval; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; IMAT, intermuscular adipose tissue; SAT, subcutaneous adipose tissue; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference.

https://doi.org/10.1371/journal.pmed.1003700.g002

traditional risk factors of T2D, HOMA-IR, or BMI and waist circumference failed to reject the null hypothesis for lack of any associations between PM density quartiles or 1-SD increment in the PM density and T2D incidence (S3 Table).

In the stratified analyses, our models showed similar associations between PM density and T2D incidence in the strata of the traditional risk factors of T2D, HOMA-IR, BMI, and waist circumference (S9 Fig).

#### Sensitivity and supplementary analyses

In the sensitivity analysis, self-reported physician-diagnosed T2D (the second criterion) was confirmed with the use of insulin or oral hypoglycemic agents or FPG  $\geq$ 126 mg/dL in the follow-up exam. Similar to the results in the main analysis, the IMAT index was associated with T2D incidence, after adjusting for the effects of traditional T2D risk factors, HOMA-IR, and clinical anthropometric indices (i.e., BMI and waist circumference, **S4 Table**).

In a supplementary analysis in participants with prediabetes at baseline, the models failed to reject the null hypothesis for lack of associations between adipose tissue biomarkers and T2D incidence (**S5 Table**).

### Discussion

In this study, we investigated the associations between CT-derived adipose tissue biomarkers and T2D incidence over a median of 6.8 years in a population-based multiethnic cohort of

normoglycemic participants. We showed that high IMAT indices were associated with T2D incidence. The association was attenuated after accounting for effects of traditional T2D risk factors, HOMA-IR, and clinical anthropometric indices (i.e., BMI and waist circumference) but remained statistically significant.

The burden attributable to T2D can be potentially avoidable through primary and secondary preventive measures in the early stages of the disease [29]. Despite the robust literature on the predictive roles of genetic, cultural, behavioral, and environmental risk factors in T2D incidence [30,31], there is a paucity of research on the association between imaging biomarkers and T2D incidence. In routine clinical practice, chest CT exams are commonly used for various cardiopulmonary indications [32,33]. Specifically, chest CT exams are well-established tools for screening for coronary artery plaques and coronary calcium scoring in participants with intermediate 10-year atherosclerotic cardiovascular disease (ASCVD) risk [13] as well as for annual screening for lung cancer in high-risk adults [14]. All these CT exams retain data on adipose depots distribution, and there is the opportunity to extract CT-derived adipose depots biomarkers from these exams.

We showed that IMAT in chest CT exams (i.e., high IMAT index) is associated with T2D incidence. The IMAT index in chest CT exams may reflect the overall (upper and lower bodies) deposition of extramyocellular adipose depots, which is associated with insulin resistance [17,34,35], and, possibly, T2D incidence. The IMAT is located close to the muscle fibers and may play an intermediary role in insulin resistance through secreting pro-inflammatory cytokines, extracellular matrix proteins, and increasing local free fatty acids, or collectively, through altering the skeletal muscle microenvironment [34,35]. High BMI (or other clinical anthropometric indices) may be cited as a confounding factor in the association between this biomarker and the T2D incidence [36,37]. However, the findings of this study showed that when the effects of these clinical anthropometric indices and the IMAT are both accounted for in the models, the IMAT index remained associated with T2D incidence.

In this study, we have demonstrated that subcutaneous adipose depots in chest CT exams (i.e., high SAT index) were associated with T2D incidence. This finding was in line with prior studies on this topic, which suggested that (upper body) subcutaneous adipose depots on the chest CT exams may be associated with adverse cardiometabolic risk factors [38]. However, SAT index correlated with BMI and obesity, and the association between this CT-derived adipose tissue biomarker and T2D incidence was (at least partly) due to the role of high BMI (the association between SAT index and T2D incidence was attenuated toward the null after adjusting for effects of BMI and waist circumference). We also did not show any associations between intramyocellular lipid content (i.e., PM density) and T2D incidence.

Compared to preceding works [8–10], in this study, the adipose tissue biomarkers were obtained from the chest CT exams that were primarily performed to assess the lung parenchymal structure. The use of these exams shows the fact that the CT-derived adipose tissue biomarkers can be extracted opportunistically from the commonly performed chest CT exams for routine cardiopulmonary clinical indications, i.e., coronary calcium scoring and lung cancer screening. Although future studies are required to confirm the findings of this study, our study builds on the recommendation of clinical guidelines on the use of chest CT exams for coronary calcium scoring and lung cancer screening and can extend the value of these CT exams. The 2018 ACC/AHA Cholesterol Guideline recommends using chest CT exams for coronary artery calcium scoring in adults without T2D for making decisions about statin therapy [39]. The CT-derived adipose tissue biomarkers can be extracted from these CT exams and may be used in recommending T2D-related preventive measures in addition to potentially making decisions about statin therapy.

This study has a few but important limitations. This population-based study was observational, and, therefore, our findings are limited by the lack of interventions to control for potential effects of residual confounders. Moreover, this study was nested within the MESA, which is primarily designed to study cardiovascular diseases and outcomes. Contrary to the majority of cardiovascular diseases, T2D diagnosis is usually made in outpatient settings, and the review of the hospital records may not be able to capture all T2D diagnoses. To address this limitation, we supplemented the review of the hospital records with self-reported physician-diagnosed T2D. Although the potential inconsistencies in the timing of the self-reported physician-diagnosed T2D may have confounded our findings, we tested the possible effects of outliers on the observed results using a sensitivity analysis. Finally, the change of CT-derived adipose tissue biomarkers over the follow-up may be a potential source of unmeasured confounding effects in this study. The trajectory of these biomarkers may provide a better understanding of their association with T2D incidence.

In conclusion, this study showed an association between IMAT at baseline and T2D incidence over the follow-up in normoglycemic participants and suggested the potential role of intermuscular adipose depots in the pathophysiology of T2D.

## Supporting information

**S1 Checklist. STROBE Statement.** STROBE, Strengthening the Reporting of Observational Studies in Epidemiology.

(DOCX)

**S1 Table. Relevant Codes of ICD-9 and ICD-10.** ICD, International Classification of Diseases.

(DOCX)

**S2 Table. Statistical analyses.** SD, standard deviation. (DOCX)

**S3 Table.** Associations of PM density and T2D incidence. Model 0: unadjusted. Model 1: adjusted for categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension. Reported *p*-values were corrected for multiple comparisons. BMI, body mass index; CI, confidence interval; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; HU, Hounsfield unit; PM, pectoralis muscle; PY, person-year; SD, standard deviation; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference. (DOCX)

S4 Table. Associations of adipose tissue biomarkers and T2D incidence (sensitivity analysis). Model 0: unadjusted. Model 1: adjusted for categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension. In this sensitivity analysis, self-reported physician-diagnosed T2D (the second criterion) was confirmed with the use of insulin or oral hypoglycemic agents or FPG  $\geq$ 126 mg/dL in the followup exam. Participants with self-reported T2D but missing information on the use of insulin or oral hypoglycemic agents or FPG in the follow-up exam (n = 17) were excluded. Reported p-values were corrected for multiple comparisons. BMI, body mass index; CI, confidence interval; FPG, fasting plasma glucose; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; HU, Hounsfield unit; IMAT, intermuscular adipose tissue; PM, pectoralis muscle; PY, person-year; SAT, subcutaneous adipose tissue; SD, standard deviation; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference. (DOCX) S5 Table. Associations of adipose tissue biomarkers and T2D incidence in participants with prediabetes (supplementary analysis). \*Models did not meet the proportional hazard assumption. Model 0: unadjusted. Model 1: adjusted for categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension. In this supplementary analysis, Cox proportional hazard models were used to study the associations between adipose tissue biomarkers and T2D incidence in participants with prediabetes at baseline. Reported *p*-values were corrected for multiple comparisons. BMI, body mass index; CI, confidence interval; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; HU, Hounsfield unit; IMAT, intermuscular adipose tissue; PM, pectoralis muscle; PY, person-year; SAT, subcutaneous adipose tissue; SD, standard deviation; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference.

(DOCX)

**S1 Fig. Flow diagram of the MESArthritis Ancillary Study.** CT, computed tomography. (TIF)

**S2 Fig. Timeline of the MESArthritis Ancillary Study.** MESA, Multi-Ethnic Study of Atherosclerosis.

(TIF)

**S3 Fig. IMAT, SAT, and PM in the chest CT exam of a participant.** The chest CT exam of a 69-year-old normoglycemic male participant is shown here. (**A**) The coronal reconstruction of the CT exam is shown. The blue dashed line indicates the slice above the superior aspect of the aortic arch. (**B**) The area within the PM with attenuation below an individualized threshold (-89 HU in this participant) was measured as the IMAT (red). (**C**) The area between the PM and skin surface was measured as the SAT (red). (**D**) The density of PM (red) was measured as the mean HU of the pixels in the PM (after removing the pixels in the IMAT). CT, computed tomography; HU, Hounsfield unit; IMAT, intermuscular adipose tissue; PM, pectoralis muscle; SAT, subcutaneous adipose tissue. (TIF)

**S4 Fig. Pattern of missing values.** In the dataset, there were missing data points in the HOMA-IR (161 [9.2%] data points), physical activity (13 [0.7%] data points), smoking status (10 [0.6%] data points), alcohol drinking status (8 [0.5%] data points), TG, HDL cholesterol, systolic and diastolic blood pressures, BMI, and waist circumference (2 [0.1%] data points in each). In this figure, the blue bars show number of participants with missing data points in the covariates marked with red circles below each bar. BMI, body mass index; HDL, high-density lipoprotein cholesterol; HOMA-IR, homeostatic model assessment–insulin resistance; TG, tri-glyceride; Waist C., waist circumference. (TIF)

**S5 Fig. Symmetric correlation matrix for adipose tissue biomarkers and traditional T2D risk factors.** The blue color is used to display the positive correlations, and the red color is used for negative correlations. The attenuation of the color is proportional to the estimated Pearson correlation coefficients (numbers in the boxes). BMI, body mass index; FPG, fasting plasma glucose; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; IMAT, intermuscular adipose tissue; PM, pectoralis muscle; SAT, subcutaneous adipose tissue; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference. (TIF)

**S6 Fig. Associations of adipose tissue biomarkers and T2D incidence.** Smooth function estimates (red lines) obtained from fitting a generalized additive Cox proportional hazard models with integrated smoothness estimation on the dataset, with estimated 95% CI (blue lines) is shown. The results are reported on the scale of the adipose tissue biomarkers (per 1-SD increment), and the models were adjusted for covariates in Model 1 (categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension). The numbers in brackets in the captions are the estimated degrees of freedom of the smooth curves. The rug marks along the x-axis indicate the adipose tissue biomarkers values. CI, confidence interval; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; IMAT, intermuscular adipose tissue; PM, pectoralis muscle; SAT, subcutaneous adipose tissue; SD, standard deviation; T2D, type 2 diabetes; TG, triglyceride.

(TIF)

**S7 Fig. Forest plot of the association of IMAT index and T2D incidence by stratification variables.** \**p*-value for interaction. Models were adjusted for covariates in Model 1 (categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension), except for the stratification variable. Reported *p*-values were corrected for multiple comparisons. BMI, body mass index; CI, confidence interval; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; IMAT, intermuscular adipose tissue; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference.

(TIF)

**S8 Fig. Forest plot of the association of SAT index and T2D incidence by stratification variables.** \**p*-value for interaction. Models were adjusted for covariates in Model 1 (categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension), except for the stratification variable. Reported *p*-values were corrected for multiple comparisons. BMI, body mass index; CI, confidence interval; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; SAT, subcutaneous adipose tissue; T2D, type 2 diabetes; TG, triglyceride; Waist C., waist circumference.

(TIF)

**S9 Fig. Forest plot of the association of PM density and T2D incidence by stratification variables.** \**p*-value for interaction. Models were adjusted for covariates in Model 1 (categorical age, sex, race/ethnicity, smoking status, alcohol drinking status, physical activity, TG, HDL cholesterol, and hypertension), except for the stratification variable. Reported *p*-values were corrected for multiple comparisons. BMI, body mass index; CI, confidence interval; HDL, high-density lipoprotein; HOMA-IR, homeostatic model assessment–insulin resistance; HR, hazard ratio; PM, pectoralis muscle; SD, standard deviation; T2D, type 2 diabetes; TG, triglyceride, Waist C., waist circumference. (TIF)

## Acknowledgments

The authors thank the other investigators, the staff, and the participants of the MESA study for their valuable contributions. A full list of participating MESA investigators and institutions can be found at MESA-NHLBI.org.

### **Author Contributions**

- **Conceptualization:** Farhad Pishgar, Mahsima Shabani, Thiago Quinaglia A. C. Silva, David A. Bluemke, Matthew Budoff, R Graham Barr, Matthew A. Allison, Alain G. Bertoni, Wendy S. Post, João A. C. Lima, Shadpour Demehri.
- Data curation: Farhad Pishgar, Shadpour Demehri.
- Formal analysis: Farhad Pishgar.
- Investigation: Farhad Pishgar, Mahsima Shabani, Thiago Quinaglia A. C. Silva, David A. Bluemke, Matthew Budoff, R Graham Barr, Matthew A. Allison, Alain G. Bertoni, Wendy S. Post, João A. C. Lima, Shadpour Demehri.
- Methodology: Farhad Pishgar, Mahsima Shabani, Thiago Quinaglia A. C. Silva, David A. Bluemke, Matthew Budoff, R Graham Barr, Matthew A. Allison, Alain G. Bertoni, Wendy S. Post, João A. C. Lima, Shadpour Demehri.
- Project administration: João A. C. Lima, Shadpour Demehri.

Resources: João A. C. Lima, Shadpour Demehri.

- Software: Farhad Pishgar, Shadpour Demehri.
- Supervision: David A. Bluemke, Matthew Budoff, R Graham Barr, Matthew A. Allison, Alain G. Bertoni, Wendy S. Post, João A. C. Lima, Shadpour Demehri.
- Validation: Farhad Pishgar, Shadpour Demehri.

Visualization: Farhad Pishgar.

- Writing original draft: Farhad Pishgar, Shadpour Demehri.
- Writing review & editing: Farhad Pishgar, Mahsima Shabani, Thiago Quinaglia A. C. Silva, David A. Bluemke, Matthew Budoff, R Graham Barr, Matthew A. Allison, Alain G. Bertoni, Wendy S. Post, João A. C. Lima, Shadpour Demehri.

#### References

- Global, Regional, and National Incidence, Prevalence, and Years Lived with Disability for 354 Diseases and Injuries for 195 Countries and Territories, 1990–2017: A Systematic Analysis for the Global Burden of Disease Study 2017. Lancet. 2018; 392(10159):1789–858. https://doi.org/10.1016/S0140-6736(18) 32279-7 PMID: 30496104.
- Global Burden of 87 Risk Factors in 204 Countries and Territories, 1990–2019: A Systematic Analysis for the Global Burden of Disease Study 2019. Lancet. 2020; 396(10258):1223–49. https://doi.org/10. 1016/S0140-6736(20)30752-2 PMID: 33069327.
- Flowers E, Lin F, Kandula NR, Allison M, Carr JJ, Ding J, et al. Body Composition and Diabetes Risk in South Asians: Findings From the MASALA and MESA Studies. Diabetes Care. 2019; 42(5):946–53. https://doi.org/10.2337/dc18-1510 PMID: 30796111.
- Chen Y, He D, Yang T, Zhou H, Xiang S, Shen L, et al. Relationship Between Body Composition Indicators and Risk of Type 2 Diabetes Mellitus in Chinese Adults. BMC Public Health. 2020; 20(1):452. https://doi.org/10.1186/s12889-020-08552-5 PMID: 32252701.
- Dale CE, Fatemifar G, Palmer TM, White J, Prieto-Merino D, Zabaneh D, et al. Causal Associations of Adiposity and Body Fat Distribution With Coronary Heart Disease, Stroke Subtypes, and Type 2 Diabetes Mellitus: A Mendelian Randomization Analysis. Circulation. 2017; 135(24):2373–88. https://doi.org/ 10.1161/circulationaha.116.026560 PMID: 28500271.
- Lotta LA, Wittemans LBL, Zuber V, Stewart ID, Sharp SJ, Luan J, et al. Association of Genetic Variants Related to Gluteofemoral vs Abdominal Fat Distribution With Type 2 Diabetes, Coronary Disease, and Cardiovascular Risk Factors. JAMA. 2018; 320(24):2553–63. https://doi.org/10.1001/jama.2018.19329 PMID: 30575882.

- Ji Y, Yiorkas AM, Frau F, Mook-Kanamori D, Staiger H, Thomas EL, et al. Genome-Wide and Abdominal MRI Data Provide Evidence That a Genetically Determined Favorable Adiposity Phenotype Is Characterized by Lower Ectopic Liver Fat and Lower Risk of Type 2 Diabetes, Heart Disease, and Hypertension. Diabetes. 2019; 68(1):207–19. https://doi.org/10.2337/db18-0708 PMID: 30352878.
- Granados A, Gebremariam A, Gidding SS, Terry JG, Carr JJ, Steffen LM, et al. Association of Abdominal Muscle Composition with Prediabetes and Diabetes: The CARDIA Study. Diabetes Obes Metab. 2019; 21(2):267–75. https://doi.org/10.1111/dom.13513 PMID: 30159995.
- Miljkovic I, Kuipers AL, Cvejkus R, Bunker CH, Patrick AL, Gordon CL, et al. Myosteatosis Increases with Aging and Is Associated with Incident Diabetes in African Ancestry Men. Obesity (Silver Spring). 2016; 24(2):476–82. https://doi.org/10.1002/oby.21328 PMID: 26694517.
- Lee JJ, Pedley A, Hoffmann U, Massaro JM, Fox CS. Association of Changes in Abdominal Fat Quantity and Quality With Incident Cardiovascular Disease Risk Factors. J Am Coll Cardiol. 2016; 68 (14):1509–21. https://doi.org/10.1016/j.jacc.2016.06.067 PMID: 27687192.
- Addison O, Marcus RL, Lastayo PC, Ryan AS. Intermuscular Fat: A Review of The Consequences and Causes. Int J Endocrinol. 2014; 2014:309570. https://doi.org/10.1155/2014/309570 PMID: 24527032.
- Brahimaj A, Rivadeneira F, Muka T, Sijbrands EJG, Franco OH, Dehghan A, et al. Novel Metabolic Indices and Incident Type 2 Diabetes Among Women and Men: The Rotterdam Study. Diabetologia. 2019; 62(9):1581–90. https://doi.org/10.1007/s00125-019-4921-2 PMID: 31183505.
- Hecht HS, Cronin P, Blaha MJ, Budoff MJ, Kazerooni EA, Narula J, et al. 2016 SCCT/STR Guidelines for Coronary Artery Calcium Scoring of Noncontrast Noncardiac Chest CT Scans: A Report of the Society of Cardiovascular Computed Tomography and Society of Thoracic Radiology. J Thorac Imaging. 2017; 32(5):W54–66. https://doi.org/10.1097/RTI.00000000000287 PMID: 28832417.
- Moyer VA. Screening for Lung Cancer: U.S. Preventive Services Task Force Recommendation Statement. Ann Intern Med. 2014; 160(5):330–8. https://doi.org/10.7326/M13-2771 PMID: 24378917.
- Mathur S, Rozenberg D, Verweel L, Orsso CE, Singer LG. Chest Computed Tomography Is a Valid Measure of Body Composition in Individuals with Advanced Lung Disease. Clin Physiol Funct Imaging. 2020; 40(5):360–8. https://doi.org/10.1111/cpf.12652 PMID: 32544296.
- McDonald M-LN, Diaz AA, Rutten E, Lutz SM, Harmouche R, San Jose Estepar R, et al. Chest Computed Tomography-derived Low Fat-free Mass Index and Mortality in COPD. Eur Respir J. 2017; 50 (6):1701134. https://doi.org/10.1183/13993003.01134-2017 PMID: 29242259.
- Boettcher M, Machann J, Stefan N, Thamer C, Häring HU, Claussen CD, et al. Intermuscular Adipose Tissue (IMAT): Association with Other Adipose Tissue Compartments and Insulin Sensitivity. J Magn Reson Imaging. 2009; 29(6):1340–5. https://doi.org/10.1002/jmri.21754 PMID: 19422021.
- Bild DE, Bluemke DA, Burke GL, Detrano R, Diez Roux AV, Folsom AR, et al. Multi-Ethnic Study of Atherosclerosis: Objectives and Design. Am J Epidemiol. 2002; 156(9):871–81. https://doi.org/10.1093/ aje/kwf113 PMID: 12397006.
- Hoffman EA, Ahmed FS, Baumhauer H, Budoff M, Carr JJ, Kronmal R, et al. Variation in The Percent of Emphysema-like Lung in a Healthy, Nonsmoking Multiethnic Sample. The MESA Lung Study. Ann Am Thorac Soc. 2014; 11(6):898–907. <u>https://doi.org/10.1513/AnnalsATS.201310-364OC</u> PMID: 24983825.
- Pishgar F, Shabani M, Silva TQAC, Bluemke DA, Budoff M, Barr RG, et al. Quantitative Analysis of Adipose Depots by Using Chest CT and Associations with All-Cause Mortality in Chronic Obstructive Pulmonary Disease: Longitudinal Analysis from MESArthritis Ancillary Study. Radiology. 2021; 299 (3):703–11. https://doi.org/10.1148/radiol.2021203959 PMID: 33825508.
- Christine PJ, Auchincloss AH, Bertoni AG, Carnethon MR, Sánchez BN, Moore K, et al. Longitudinal Associations Between Neighborhood Physical and Social Environments and Incident Type 2 Diabetes Mellitus: The Multi-Ethnic Study of Atherosclerosis (MESA). JAMA Intern Med. 2015; 175(8):1311–20. https://doi.org/10.1001/jamainternmed.2015.2691 PMID: 26121402.
- Mackey RH, Greenland P, Goff DC Jr., Lloyd-Jones D, Sibley CT, Mora S. High-density Lipoprotein Cholesterol and Particle Concentrations, Carotid Atherosclerosis, and Coronary Events: MESA (Multi-Ethnic Study of Atherosclerosis). J Am Coll Cardiol. 2012; 60(6):508–16. <u>https://doi.org/10.1016/j.jacc.</u> 2012.03.060 PMID: 22796256.
- Couper D, LaVange LM, Han M, Barr RG, Bleecker E, Hoffman EA, et al. Design of the Subpopulations and Intermediate Outcomes in COPD Study (SPIROMICS). Thorax. 2014; 69(5):491–4. <u>https://doi.org/ 10.1136/thoraxjnl-2013-203897 PMID: 24029743</u>.
- McDonald ML, Diaz AA, Ross JC, San Jose Estepar R, Zhou L, Regan EA, et al. Quantitative Computed Tomography Measures of Pectoralis Muscle Area and Disease Severity in Chronic Obstructive Pulmonary Disease. A Cross-sectional Study. Ann Am Thorac Soc 2014; 11(3):326–34. https://doi.org/ 10.1513/AnnalsATS.201307-229OC PMID: 24558953.

- 25. Mühlberg A, Museyko O, Bousson V, Pottecher P, Laredo J-D, Engelke K. Three-dimensional Distribution of Muscle and Adipose Tissue of the Thigh at CT: Association with Acute Hip Fracture. Radiology. 2019; 290(2):426–34. https://doi.org/10.1148/radiol.2018181112 PMID: 30457478.
- Grundy SM, Brewer HB, Cleeman JI, Smith SC, Lenfant C. Definition of Metabolic Syndrome. Circulation. 2004; 109(3):433–8. https://doi.org/10.1161/01.CIR.0000111245.75752.C6 PMID: 14744958.
- van Buuren S, Groothuis-Oudshoorn K. mice: Multivariate Imputation by Chained Equations in R. J Stat Softw 2011; 45(3):1–67. https://doi.org/10.18637/jss.v045.i03
- Benjamini Y, Hochberg Y. Controlling the False Discovery Rate: A Practical and Powerful Approach to Multiple Testing. J R Stat Soc Series B Stat Methodol. 1995; 57(1):289–300. https://doi.org/10.1111/j. 2517-6161.1995.tb02031.x
- Li R, Zhang P, Barker LE, Chowdhury FM, Zhang X. Cost-effectiveness of Interventions to Prevent and Control Diabetes Mellitus: a Systematic Review. Diabetes Care. 2010; 33(8):1872–94. <u>https://doi.org/ 10.2337/dc10-0843</u> PMID: 20668156.
- Ardisson Korat AV, Willett WC, Hu FB. Diet, Lifestyle, and Genetic Risk Factors for Type 2 Diabetes: A Review from the Nurses' Health Study, Nurses' Health Study 2, and Health Professionals' Follow-up Study. Curr Nutr Rep. 2014; 3(4):345–54. <u>https://doi.org/10.1007/s13668-014-0103-5</u> PMID: 25599007.
- Bellou V, Belbasis L, Tzoulaki I, Evangelou E. Risk Factors for Type 2 Diabetes Mellitus: An Exposurewide Umbrella Review of Meta-analyses. PLoS ONE. 2018; 13(3):e0194127. https://doi.org/10.1371/ journal.pone.0194127 PMID: 29558518.
- Mullan CP, Madan R, Trotman-Dickenson B, Qian X, Jacobson FL, Hunsaker A. Radiology of Chest Wall Masses. AJR Am J Roentgenol. 2011; 197(3):W460–70. https://doi.org/10.2214/AJR.10.7259 PMID: 21862774.
- Raju S, Ghosh S, Mehta AC. Chest CT Signs in Pulmonary Disease: A Pictorial Review. Chest. 2017; 151(6):1356–74. https://doi.org/10.1016/j.chest.2016.12.033 PMID: 28212835.
- Khan IM, Perrard XY, Brunner G, Lui H, Sparks LM, Smith SR, et al. Intermuscular and Perimuscular Fat Expansion in Obesity Correlates with Skeletal Muscle T cell and Macrophage Infiltration and Insulin Resistance. Int J Obes (Lond). 2015; 39(11):1607–18. https://doi.org/10.1038/ijo.2015.104 PMID: 26041698.
- Sachs S, Zarini S, Kahn DE, Harrison KA, Perreault L, Phang T, et al. Intermuscular Adipose Tissue Directly Modulates Skeletal Muscle Insulin Sensitivity in Humans. Am J Physiol Endocrinol Metab. 2019; 316(5):E866–79. https://doi.org/10.1152/ajpendo.00243.2018 PMID: 30620635.
- 36. Goodpaster BH, Krishnaswami S, Resnick H, Kelley DE, Haggerty C, Harris TB, et al. Association Between Regional Adipose Tissue Distribution and Both Type 2 Diabetes and Impaired Glucose Tolerance in Elderly Men and Women. Diabetes Care. 2003; 26(2):372–9. https://doi.org/10.2337/diacare. 26.2.372 PMID: 12547865.
- Albu JB, Kovera AJ, Allen L, Wainwright M, Berk E, Raja-Khan N, et al. Independent Association of Insulin Resistance With Larger Amounts of Intermuscular Adipose Tissue and a Greater Acute Insulin Response to Glucose in African American Than in White Nondiabetic Women. Am J Clin Nutr. 2005; 82 (6):1210–7. https://doi.org/10.1093/ajcn/82.6.1210 PMID: 16332653.
- Lee JJ, Pedley A, Therkelsen KE, Hoffmann U, Massaro JM, Levy D, et al. Upper Body Subcutaneous Fat Is Associated with Cardiometabolic Risk Factors. Am J Med. 2017; 130(8):958–66.e1. https://doi. org/10.1016/j.amjmed.2017.01.044 PMID: 28238696.
- 39. Grundy SM, Stone NJ, Bailey AL, Beam C, Birtcher KK, Blumenthal RS, et al. 2018 AHA/ACC/ AACVPR/AAPA/ABC/ACPM/ADA/AGS/APhA/ASPC/NLA/PCNA Guideline on the Management of Blood Cholesterol: A Report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines. Circulation. 2019; 139(25):e1082–143. https://doi.org/10.1161/ CIR.00000000000625 PMID: 30586774.